Response to Reviewer-1
Thank you for your high evaluation of my article. The parts that you pointed out have been added or changed as described below. Thank you for your cooperation.

Major issues
As you pointed out, including the case of recurrence, there are 4 patients in Stage IV. This is a misleading and confusing description. In the revised version, we decided to focus on two patients, Case 2 and Case 3, which were in Stage IV at the first onset.

Minor issues
1) Page 10 The first ‘the’ in the sentence was deleted.
2) Page 6 The article you mentioned has been added to the manuscript and literature.

Response to Reviewer-2
Thank you for your high evaluation of my article. The part pointed out by Reviewer-1 has been added and changed a little. Thank you for your cooperation.